Author’s response to reviews

Title: Trends in Comprehensive Abortion Care (CAC) and Characteristics of Women Receiving abortion care in Tertiary Hospital in Nepal.

Authors:
Jamuna Sayami (jamunats10@gmail.com)

Version: 2 Date: 12 Sep 2017

Author’s response to reviews:
Revision instructions have been followed.